Author's response to reviews

Title: A pilot study of Aboriginal health promotion from an ecological perspective

Authors:

Rachel E Reilly (rachel@unimelb.edu.au)
Marion Cincotta (marionc@vaccho.com.au)
Joyce Doyle (jdoyle@bigpond.com)
Bradley R Firebrace (bradleyf@unimelb.edu.au)
Margaret Cargo (margaret.cargo@unisa.edu.au)
Gemma van den Tol (gvandentol@gvct.com.au)
Denise Morgan-Bulled (denise@vineymorganams.com.au)
Kevin G Rowley (rowleyk@unimelb.edu.au)

Version: 5 Date: 15 August 2011

Author's response to reviews:

Dear Editorial Production Team,

Thank you for feedback. We have made the following changes in response to your requests:

Major Revisions:

1. Author 7 is listed as Denise Morgan-Bulled in both the admission system and on the manuscript.

2. All tables have been re-formatted using a tables tool in Word.

3. Key-words have been removed from the admission system to make it consistent with the manuscript.

Minor revisions:

4. The names of the Heart Health Project Steering committee have been moved the the acknowledgements section. We request that 'for the Heart Health Project Steering Committee' remains in the list of authors because we consider this the appropriate recognition of the expert contribution of these senior Aboriginal community members. We have added a brief explanation of this to the 'Authors contributions' section.

5. The figure legend is now after the references.

6. A number of inappropriate 'Ins' inserted by the Endnote system have been removed from the reference list, and missing issue numbers have been inserted.
Many thanks,

Rachel Reilly